Author's response to reviews

Title: Effects of MTHFR gene polymorphism on the clinical and electrophysiological characteristics of migraine

Authors:

Julia E Azimova (azimova.j@mail.ru)
Alexey V Sergeev (sergeev.amd@gmail.com)
Liubov A Korobeynikova (Korobeynikova_L@mail.ru)
Gadjii O Shaikhaev (shaikhaev@gmail.com)
Eugene E Klimov (klimov_eugeney@mail.ru)
Gyusal R Tabeeva (grtabeeva@gmail.com)
Natalia S Kondratieva (natalia_kondratieva@mail.ru)
Zarema G Kokaeva (zaremak@inbox.ru)
Natalia M Fokina (nfokina@mma.ru)
Kirill V Skorobogatykh (post.kirill@gmail.com)

Version: 3 Date: 27 June 2013

Author's response to reviews: see over
Dear Editors,

We submit here our second revised MS entitled "Effects of MTHFR gene polymorphism on the clinical and electrophysiological characteristics of migraine" by Aimova et al. for publication in *BMC Neurology*.

**Response to the reviewers’ reports**

Thank you very much for your reviewers’ reports which was very useful for us. We revised our manuscript according your comments.

Allow us to response to the comments.

**To comments of Dr Elisa Rubino review:**

We thank the referee for carefully reviewing our paper.

**To comments of Dr Gretchen E Tietjen review:**

We thank the referee for carefully reviewing our paper.

We have introduced amendments into the section «Patients and methods» of the manuscript. The control group was composed of our friends and co-worker from Institutes and Universities, who was not diagnosis of migraine or other type of headache.

The misprints and stylistically awkward phrasing also are corrected, and English of the article is edited and checked out by our English speaking co-worker from USA.

Yours sincerely,

Eugene Klimov.